Author’s response to reviews

Title: Additive interaction of snoring and body mass index on the prevalence of Metabolic Syndrome among Chinese coal mine employees: A Cross-Sectional Study

Authors:

Yanyan Li (liyanyan_sxmu@sina.com)
Qian Gao (gaoqian1206@163.com)
Lu Li (Qian.Lu93@163.com)
Yanan Shen (SYNPEKING@163.com)
Qing Lu (qlu@epi.msu.edu)
Jianjun Huang (hjj1958@126.com)
Chenming Sun (bun005@126.com)
Hui Wang (annieyuner@163.com)
Nan Qiao (qiaonan_sxmu@sina.com)
Cong Wang (wangcong_sxmu@sina.com)
Haixia Zhang (zhx.196@163.com)
Tong Wang (tongwang@sxmu.edu.cn)

Version: 2 Date: 07 Feb 2019

Author’s response to reviews:

Dear Editor and Reviewers,

Thank you for the valuable comments on the manuscript entitled "Additive interaction of snoring and body mass index on the prevalence of Metabolic Syndrome: A Cross-Sectional Study" (ID: BEND-D-17-00304R1). Those comments are very helpful to us. We have studied these comments carefully and revised accordingly. Responses to the comments are as follows:

1 Editor Comments:
Please have a native English speaker check though your manuscript as there are still some language issues, or use a professional language service. Thank you.

Response: Thank you for your comments.

A native English speaker read the paper and helped us to correct the grammar errors.

2 Reviewer comments: The authors addressed all concerns of both reviews. The authors explained clearly how they addressed the concerns and the changes made to the text of the manuscript. There remain several minor English language issues. For example, in the Introduction there are sentences that require use of definite articles and use of plural."

Response: Thank you for your comments.

We have corrected the English language issues (For example: Introduction section, page 3, line 3; Introduction section, page 5, line 1; Method section, page 7, line 29)